Author’s response to reviews

Title: Title Prison health situation and health rights of young people incarcerated in sub-Saharan African prisons and detention centres: A scoping review of extant literature.

Authors:

Marie Claire Van Hout (m.c.vanhout@ljmu.ac.uk)

Rosemary Mhlanga-Gunda (rosemhlanga3@gmail.com)

Version: 1 Date: 17 Mar 2019

Author’s response to reviews:

Dear Editor,

Please find uploaded the revised and clean manuscript, tables, figure and supplemental table. I have emailed you a marked copy so you can see that all of your edits were done.

Sincerely,

Prof Marie Claire Van Hout